Author's response to reviews

Title: Dextrocardia and asplenia in situs inversus totalis in an infant: a case report.

Authors:

   ABNISH KUMAR Jr-III (abnish23@gmail.com)
   MANOJ KUMAR SINGH MD (drmanuped@yahoo.com)
   NEERAJ YADAV MD (neeraiyadu@rediffmail.com)

Version: 2 Date: 24 March 2014

Author's response to reviews: see over
FROM

DR ABNISH KUMAR

DEPARTMENT OF PAEDIATRICS

S N MEDICAL COLLEGE, AGRA

TO

THE EDITOR

Journal of medical case reports

SIR

REF: TITLE- DEXTROCARDIA AND ASPLENA IN SITUS INVERSUS TOTALIS IN AN INFANT.

TYPE- A CASE REPORT

In reference to the above title, I as a corresponding author, submit the manuscript for publication in Journal name [Journal of medical case reports]. Few cases were reported earlier on this topic. Our study have a broader clinical impact across more than one area of medicine. It will significantly advance our understanding about situs inversus. This manuscript has not been published or considered for publication by any other journal elsewhere. Kindly consider the manuscript for publication in your journal.

Thank you

Signature – abnish kumar

Corresponding author name

ABNISH KUMAR